Author’s response to reviews

Title: CHILD HEALTH, HOUSEHOLD ENVIRONMENT, TEMPERATURE AND RAINFALL ANOMALIES IN HONDURAS: A SOCIO-CLIMATE DATA LINKED ANALYSIS

Authors:

Cristina Bradatan (cristina.bradatan@ttu.edu)
Jeff Dennis (jeff.dennis@ttuhsc.edu)
Nadia Flores Yeffal (nadia.flores@ttu.edu)
Sharmistha Swain (sharmistha.swain@ttu.edu)

Version: 3 Date: 26 Nov 2019

Author’s response to reviews:

Dear Editor,

thank you for your comments on our paper. A revised version of our paper has been uploaded. Please find below how we responded to these issues:

1. Although subheadings are appreciated in the Methods section, a subheading is also named Methods, and it seems that the structure needs to be revised. Also, note that manuscript sections should not be numbered.

- We fixed these issues.

2. The Discussion consists of a single paragraph that does not reflect the proper consideration of the evidence and the perspective. In contrast, the Conclusions, which should consist of a single, crisp paragraph that summarize the findings, is way too long.

- We changed the structure of the Discussion and Conclusions. We reduced Conclusions to one phrase.

3. Finally, the references must be numbered and listed in accordance with the instructions.

- We fixed these.

Cristina Bradatan